Author’s response to reviews

Title: Acute Versus Chronic Methotrexate Poisoning; a Retrospective Study

Authors:

Arman Ahmadzadeh (arman319@yahoo.com)

Nasim Zamani (nasim.zamani@gmail.com)

Hossein Hassanian-Moghaddam (hassanian@sbmu.ac.ir)

Seyed Kaveh Hadeiy (kaveh.hadee@yahoo.com)

Parinaz Parhizgar (p.parhizgar@yahoo.com)

Version: 2 Date: 17 May 2019

Author’s response to reviews:

Dear Prof William Spenceley Jonesaa

BMC Pharmacology and Toxicology

Please find our response to editor’s comment in which we have revised suggested parts. Editor Comments:

1. Please remove the Acknowledgement and Authors' Contribution from the Title page of your manuscript.

RE: DONE

2. In the “Ethics approval and consent to participate” section of the Declarations please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

RE: DONE

3. In your “Ethics approval and consent to participate” section of your Declarations, please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state it in this section. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

RE: DONE

4. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

RE: DONE

5. We have noted that author Arman Ahmadzadeh, Seyed Kaveh Hadeiy, Parinaz Parhizgar are missing in the listed authors’ contributions. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance
and criteria for authorship can be found here:
http://www.biomedcentral.com/submissions/editorial-policies#authorship

RE: CHANGED and DONE 6. Please clarify who the initials of MR, HZ, HB, LG are referring to. If you need to add additional authors to your author list you will need to fill out a change in authorship form and send it to the Editorial office to be approved by the Editor. The form can be found here: https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5.

RE: There is no new co-author, a simple mistake from my side. Initials of co-author are put in authors contribution. 7. Thank you for uploading your authorship form and response to reviewers. As these files are no longer required at this stage of the editorial process, please remove them from your submission.

RE: DONE 8. Please upload your revised manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. Please also ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

RE: DONE

Yours truly, Hossein Hassanian-Moghaddam MD, FACMT

Corresponding Author